Author's response to reviews

Title: The UK Clinical Research Network - has it been a success for dermatology clinical trials?

Authors:

Kim S Thomas (kim.thomas@nottingham.ac.uk)
Karin Koller (Karin.koller@ntlworld.com)
Katharine Foster (Katharine.foster@nottingham.ac.uk)
Jo Perdue (jo.perdue@nottingham.ac.uk)
Lisa Charlesworth (lisa.charlesworth@nottingham.ac.uk)
Joanne R Chalmers (joanne.chalmers@Nottingham.ac.uk)

Version: 4 Date: 15 June 2011

Author's response to reviews:

Please accept this latest version of the main paper (a couple of typographical errors have been corrected) and a revised Figure 2 (the blue and red stars were not present on the uploaded version)

Thanks very much.

Kim Thomas